This article was downloaded by: [OCDE - Library & Archives] On: 05 December 2013, At: 01:13 Publisher: Routledge Informa Ltd Registered in England and Wales Registered Number: 1072954 Registered office: Mortimer House, 37-41 Mortimer Street, London W1T 3JH, UK



Economic Systems Research

Publication details, including instructions for authors and subscription information: <u>http://www.tandfonline.com/loi/cesr20</u>

THE CONSTRUCTION OF WORLD INPUT-OUTPUT TABLES IN THE WIOD PROJECT

Erik Dietzenbacher^a, Bart Los^a, Robert Stehrer^b, Marcel Timmer^a & Gaaitzen de Vries^a

 $^{\rm a}$ Faculty of Economics and Business , University of Groningen , Groningen , The Netherlands

^b Vienna Institute for International Economic Studies (wiiw), Vienna, Austria

Published online: 21 Mar 2013.

To cite this article: Erik Dietzenbacher , Bart Los , Robert Stehrer , Marcel Timmer & Gaaitzen de Vries (2013) THE CONSTRUCTION OF WORLD INPUT-OUTPUT TABLES IN THE WIOD PROJECT, Economic Systems Research, 25:1, 71-98, DOI: <u>10.1080/09535314.2012.761180</u>

To link to this article: <u>http://dx.doi.org/10.1080/09535314.2012.761180</u>

PLEASE SCROLL DOWN FOR ARTICLE

Taylor & Francis makes every effort to ensure the accuracy of all the information (the "Content") contained in the publications on our platform. However, Taylor & Francis, our agents, and our licensors make no representations or warranties whatsoever as to the accuracy, completeness, or suitability for any purpose of the Content. Any opinions and views expressed in this publication are the opinions and views of the authors, and are not the views of or endorsed by Taylor & Francis. The accuracy of the Content should not be relied upon and should be independently verified with primary sources of information. Taylor and Francis shall not be liable for any losses, actions, claims, proceedings, demands, costs, expenses, damages, and other liabilities whatsoever or howsoever caused arising directly or indirectly in connection with, in relation to or arising out of the use of the Content.

This article may be used for research, teaching, and private study purposes. Any substantial or systematic reproduction, redistribution, reselling, loan, sub-licensing, systematic supply, or distribution in any form to anyone is expressly forbidden. Terms &

Conditions of access and use can be found at <u>http://www.tandfonline.com/page/terms-and-conditions</u>

THE CONSTRUCTION OF WORLD INPUT–OUTPUT TABLES IN THE WIOD PROJECT

ERIK DIETZENBACHER^a*, BART LOS^a, ROBERT STEHRER^b, MARCEL TIMMER^a and GAAITZEN DE VRIES^a

^a Faculty of Economics and Business, University of Groningen, Groningen, The Netherlands; ^bVienna Institute for International Economic Studies (wiiw), Vienna, Austria

(Received 18 August 2012; In final form 18 December 2012)

This article describes the construction of the World Input–Output Tables (WIOTs) that constitute the core of the World Input–Output Database. WIOTs are available for the period 1995–2009 and give the values of transactions among 35 industries in 40 countries plus the 'Rest of the World' and from these industries to households, governments and users of capital goods in the same set of countries. The article describes how information from the National Accounts, Supply and Use Tables and International Trade Statistics have been harmonized, reconciled and used for estimation procedures to arrive at a consistent time series of WIOTs.

Keywords: World input-output tables; Supply and use tables; Bilateral trade data

1. INTRODUCTION

According to official trade statistics, China was the largest global exporter in 2012, having surpassed both Germany and the USA. In 1999, just 13 years ago, Chinese exports were less than a third of American exports (*The Economist*, 2012). To what extent did Chinese income and employment benefit from this surge in exports? To what extent did it suffer from increasing pollution associated with producing goods for foreign markets? And to what extent have countries that are important suppliers of China's imports such as Australia, Japan and Korea, gained from Chinese growth? More in general, what are the effects of fragmentation of production on the distribution of employment, income and the generation of emissions across countries and industries?

These are very relevant questions, given that many of today's products and services are no longer produced within a single country. Whereas, many electronic products are labelled 'assembled in China', their key components are often produced also in other parts of the world. In the last couple of decades, production processes have been sliced up more and more into ever smaller parts, many of which are subcontracted to specialized suppliers. In addition, this fragmentation crosses the borders of countries more and more. This coincides with the common viewpoint today that products and services are made in global value chains.¹

^{*}Corresponding author. E-mail: h.w.a.dietzenbacher@rug.nl

¹ In the same vein, the OECD and the WTO launched the 'Made in the World' initiative and proposed 'trade in value added' as a better approach for the measurement for international trade (OECD–WTO, 2012).

However, as pointed out by the European Commissioner for Trade, Karel De Gucht:²

'... though we are aware of the rising importance of global value chains, we have so far been unable to properly measure their size, nature and effect. This is because our current statistical apparatus does not capture the domestic activity contained in a traded good or service'.

Recently, economic theory has also paid attention to the causes and consequences of this process which Baldwin (2006) refers to as the 'second wave of unbundling'. Grossman and Rossi-Hansberg (2008), for example, constructed a theory that does not focus on trade in products, but on 'trade in tasks'. The tasks are implicitly related to the production of intermediate inputs and vary with respect to the intensities of labour inputs of different skill levels. Firms decide on the location of these tasks on the basis of cost comparisons. As a consequence, some intermediate inputs might be produced in the home country, while others might be relocated abroad (either by acquiring or building production facilities, i.e. by foreign direct investment (FDI), or by buying from foreign suppliers). Grossman and Rossi-Hansberg (2008) show that improvements in the opportunities to relocate tasks might have an effect on the host country that is comparable to that of factor-augmenting technological change. Their theory leads to conclusions regarding the effects of trade that differ from what can be concluded from traditional trade theories. In the concluding paragraph, the authors state that 'the globalization of production processes mandates a new approach to trade data collection, one that records international transactions, much like domestic transactions have been recorded for many years' (Grossman and Rossi-Hansberg, 2008, p. 1996). Without such data, we are groping in the dark about the causes and consequences of the changing nature of international trade.³

Production processes are characterized by international fragmentation leading to an interdependent structure which has to be accounted for. The data that provide a description of such an interdependent production structure are given in supply and use tables (SUTs) and/or input–output tables (IOTs). Given the on-going trend in globalization, we agree with Wiedmann et al. (2011) that a database that is useful for analysing economic, social and environmental issues and policies should take *each* of the following three aspects into account. It must (i) be *global*, (ii) cover *changes over time* in order to evaluate past developments, and (iii) include a variety of *socio-economic and environmental indicators*. Moreover, it is necessary to have all data in a coherent framework (e.g. using the same product and industry classifications and consistent definitions).

The World Input–Output Database (WIOD) project (WIOD: Construction and Applications) was set up to create such an all-encompassing database, which provides a tool that can address both the quest for indicators by policy-makers and the need for empirical observations to test and quantify theories by academic researchers. The project ran for 3 years (May 2009–April 2012) and 11 international partners were involved.^{4,5} The database

² Available at: http://trade.ec.europa.eu/doclib/docs/2012/april/tradoc_149337.pdf.

³ An alternative theory of emerging geographical patterns in internationally fragmented production processes has been proposed by Costinot et al. (forthcoming).

⁴ More information about the WIOD project can be found at: http://www.wiod.org/index.htm.

⁵ The participants were: University of Groningen, The Netherlands; Institute for Prospective Technological Studies, Seville, Spain; The Vienna Institute for International Economic Studies (wiiw), Vienna, Austria; Zentrum

allows for addressing issues related to fragmentation and socio-economic aspects (such as jobs or the creation of value added) as well as environmental aspects (such as energy use, various emissions to air, or the use of water). The database combines detailed information on national production activities and international trade data. For each country, tables are constructed that reflect how much of each of 59 products is produced and used by each of 35 industries. By linking these tables to trade data, it is estimated, for example, how many dollars of Belgian fabricated metal products are used by the French transport equipment industry. This type of information is available in the WIOD database for 40 countries (all 27 European Union (EU) countries and 13 major other countries), plus estimates for the rest of the world (RoW); for the time period 1995–2007 (and estimates for 2008 and 2009); in current basic prices and in basic prices of the previous year. It should be emphasized that all data in the WIOD are obtained from official national statistics and are consistent with the National Accounts. The full database is publicly and free of charge available at: http://www.wiod.org/database/index.htm.

The rest of this paper is structured as follows. Section 2 will be devoted to the principles underlying the database, i.e. the choices that have been made in its construction. As a comparison between the various contributions in this issue reveals, different groups of researchers made different choices, as a consequence of which our world input-output tables (WIOTs) differ from the multi-regional input-output (MRIO) tables constructed by others.⁶ A careful inspection of these principles (next to the coverage of the WIOTs in terms of industries, countries and years) should help the user to make a choice between our WIOTs and the available alternatives. The description of the construction of the WIOTs distinguishes three steps.⁷ Section 3 discusses the construction of the time series of national SUTs and pays attention to aspects of harmonization and benchmarking the SUTs on the National Accounts. Section 4 describes how the international SUTs are derived. An important aspect is the estimation of the imports in the use table for which a bilateral trade database was built. Section 5 details how the international SUTs where combined so as to yield a world SUT, which then was transformed into a WIOT. Section 6 gives a listing of the contents of the WIOD database, including the socio-economic and environmental satellite accounts. The last section concludes.

2. THE UNDERLYING PRINCIPLES AND CHOICES

Constructing a large database like in the WIOD project implies that several choices need to be made. These choices are often directed by the particular applications the constructors have in mind when designing the database and its underlying fundamental principles.

für Europäische Wirtschaftsforschung, Mannheim, Germany; Österreichisches Institut für Wirtschaftsforschung, Vienna, Austria; Hochschule Konstanz, Germany; The Conference Board Europe, Brussels, Belgium; CPB Netherlands Bureau for Economic Policy Analysis, The Hague, The Netherlands; Institute of Communication and Computer Systems, Athens, Greece; Central Recherche SA, Paris, France; OECD, Paris, France.

⁶ In the environmental and ecological literature, our WIOTs are often termed MRIO tables (or MRIOs). We prefer not to use the term MRIO, because of the traditional conceptual distinction between multi-regional IOTs and interregional IOTs (Miller and Blair, 2009) and because of the fact that our tables do not provide detail at the subnational level.

⁷A detailed description of the contents, sources and methods for the WIOD database can be found in Timmer (ed., 2012).

E. DIETZENBACHER et al.

Uncovering these is important in order to understand the differences between various alternative databases.⁸ Consequently, one database should *not* be seen as 'better' than another database. It may be better (or more appropriate) for answering some questions but not for other questions. For example, it must be a deliberate choice to include as many countries as possible or to strive for a high level of reliability by limiting the number of countries (knowing that for certain countries the quality of the data is poor or that data are not official). Another example is the choice of how to deal with discrepancies that exist between the export and import values recorded in the National Accounts Statistics (NAS) and in the International Trade Statistics (ITS), including the well-known problem that mirror statistics between bilateral trade partners are far from consistent.⁹ One option is to take the absolute values from NAS and to assign these to countries-of-origin and countries-of-destination, using shares obtained from ITS. Other options are to take the values from ITS and adapt the product-level exports and imports from NAS, or to develop a construction algorithm that allows for specific choices at product or industry level. The rest of this section will point out the choices that have been made in constructing the WIOD database. In so doing, it will also sketch the main construction steps, each of which will be described in detail in later sections (references to related literature can be found in the sections that deal with the details).

First, we have used national SUTs as the starting point for the construction (Section 3.1). SUTs are the core statistical sources from which National Statistical Institutes derive national IOTs. In IOTs, it is assumed that each industry produces goods and services in exactly one product class. Consequently, the distinction between industry and product vanishes and the tables become square (or, in statistical parlance, symmetric). SUTs on the other hand are usually non-square and allow for secondary production, which makes them better reflect 'reality'. The supply table provides information on how much of each product is produced by each domestic industry and how much of this product is imported. The use table indicates the use of each product (combining domestically produced and imported products) by each of the industries and final use categories (e.g. consumption by households and government, investments, and gross exports). Both supply tables and use tables are thus of the product-by-industry dimension. Therefore, linking SUTs with international trade data (which are product based) and with socio-economic and environmental data (which are mainly industry-based) becomes more straightforward and accurate (Section 4.2).

Second, we have used the National Accounts as the benchmark. Typically, SUTs are only available for a limited set of years (e.g. every 5 years) and once released by the national statistical institute revisions are rare. National Accounts on the other hand are usually revised several times. This is because statistical systems develop, new methodologies and accounting rules are used, classification schemes change and new data become available. Occasionally, revisions are also carried through to ensure consistency and comparability over time. These revisions can be substantial, especially at a detailed industry level, implying discrepancies between information from the latest version of the National Accounts for a

⁸ See, for example, Andrew and Peters (2013), Lenzen et al. (2013), Meng et al. (2013) and Tukker et al. (2013), all of which are included in this special issue. See also Oosterhaven et al. (2008) on the Asian International IOTs project, Wang et al. (2010) on a project at the US International Trade Commission and Wiebe et al. (2012) on the Global Resource Accounting Model project.

⁹ See the discussion in Guo et al. (2009).

certain year and the published SUT for that year. So, in our approach, any revision of the National Accounts leads to an adaptation of the benchmark national SUTs to make them match (see Section 3.2 for details).

Third, one of our aims was to arrive at a time series of WIOTs. This required us to devise a method to estimate national SUTs for non-benchmark years as well. Time series for (gross) output and value added by industry, total imports and total exports and final use by use category were taken from the NAS and were used as constraints when generating time series of SUTs with the so-called SUT-RAS method. This method was developed for this specific purpose. Time series of SUTs have been derived for two price concepts: basic prices and purchasers' prices. Basic prices reflect all costs borne by the producer, whereas purchasers' prices reflect the amounts paid by the purchaser. Supply tables are always in basic prices and often have additional information on margins and net taxes by product. Use tables as available from public data sources are typically in purchasers' prices and had to be transformed to basic prices within the construction procedures. The difference between the two types of use tables is given in the so-called valuation matrices with the trade and transportation margins and the net taxes, which had to be estimated (Section 3.2 provides details about the harmonization of price concepts and derivation of the valuation matrices, while Section 3.3 describes the construction of time series of national SUTs by means of SUT-RAS).

Fourth, within the WIOD, the choice was made that the data used in the project should be publicly available. This ensures that users of the WIOTs are able to trace the construction process and are able to derive alternative tables by making different sets of assumptions. Moreover, officially published data are more reliable because thorough checking and validation procedures have been adopted by the National Statistical Institutes (when compared to data generated on an *ad hoc* basis for specific research purposes).

Fifth, an improved allocation of imports of goods has been applied. In the process of construction, the national SUTs have been combined with information from ITS to construct what we call international SUTs. Recall that use tables include both domestically produced and imported products. They have been split into the use of domestic products and use of foreign products first, and in a second stage the use of foreign products was split according to country of origin. The standard assumption in most databases is to apply import proportionality (where the same, fixed percentage of total use of a product is assumed to be imported, irrespective of its purchaser). For the import of goods, we have developed an estimation method that does not rely on this standard import proportionality assumption. Instead, we have determined (for each product) the share of its imports that goes to the 'intermediate consumption', to 'final consumption', and to the 'gross fixed capital formation' (the so-called end-use categories). This distinction by end-use categories was based on a refinement of the well-known BEC codes ('broad economic categories') distinguishing detailed trade data by use. Within each end-use category, the allocation was (as dictated by a lack of additional information) based on the proportionality assumption.¹⁰ For intermediate use by industries, for example, we had to apply ratios between imported use and total use

¹⁰ Personal communication with National Statistical Institutes led us to the conclusion that procedures to arrive at imports in published use tables and IOTs are often based on import proportionality at lower levels of product aggregation. Tables constructed along these lines do, therefore, not convey data of higher quality. Only for a small number of countries and years, import tables have been constructed using firm-level survey data on the sourcing of products.

that were equal across industries, but differed from the corresponding ratio for consumption purposes. A similar procedure was used to split the imports table according to the country of origin. Unlike under the standard proportionality assumption, country import shares differ across end-use categories (but not within these categories). Section 4.1 provides details and references to the related literature.

Sixth, given the types of application that we had in mind for the WIOD database, it is important also to have detailed information on the trade in services. For services trade, however, no standardized database on bilateral flows exists. The data have been collected from various sources (including UN, OECD, Eurostat), checked for consistency and integrated into a bilateral services trade database (see the last part of Section 4.1 for the particularities).

Seventh, for some applications, it is important to have data in constant prices. Therefore, WIOTs in pyp have been constructed based on gross output deflators from the National Accounts of each country, implicitly deflating imports by the exporters' gross output deflators (details are described in the last part of Section 5.2).

Eighth, one of the aims in the WIOD project was to link the WIOTs to satellite accounts that provide data at the same industry level. The socio-economic accounts focus on the inputs of production factors and the environmental accounts list requirements for and effects of production (e.g. energy use and emissions, respectively). In this paper, we do not describe these satellite accounts. See Erumban et al. (2012a) and Genty et al. (2012) for descriptions of the socio-economic and environmental satellite accounts, respectively.

These underlying principles and choices have guided us through the process of constructing the WIOD database. The next sections will go into the details of constructing a WIOT. First, we will discuss the construction of a time series of national SUTs, after which we will describe how they were linked to trade data so as to yield international SUTs. Finally, we show how international SUTs for the 40 countries were used to produce a world SUT. Finally, the world SUT was then transformed into a WIOT with additional modelling of the so-called RoW (which covers countries for which no national SUTs have been derived).

3. CONSTRUCTION OF A TIME SERIES OF NATIONAL SUTS

In the construction of the time series of national SUTs, we started with the SUTs as published by the National Statistical Institutes. Because national SUTs are only infrequently available and are often not harmonized over time, they have been benchmarked on consistent time series from the National Accounts. The construction involved three steps: harmonization and standardization of the published SUTs across countries and time (Section 3.1), benchmarking the harmonized national SUTs to National Accounts (Section 3.2) and building a time series of national SUTs (Section 3.3).

3.1. Harmonization and Standardization of the Published SUTs

The national SUTs were derived from statistics published by the National Statistical Institutes. Although SUTs and national account statistics are increasingly harmonized across countries, differences still remain, in particular for less developed countries. The national data also differ in the level of product and industry detail provided. Harmonization was thus required and involved the following aspects (details on a country-by-country basis are reported in Erumban et al., 2012b).

Commodity-by-industry classification: The level of industry and commodity detail in the basic SUTs and IOTs varied widely across countries, variables and periods. National SUTs (or IOTs) were converted into tables with 59 products (based on the international classification of products by activity, CPA) and 35 industries (based on the NACE revision 1 which corresponds to ISIC revision 3). For this, we developed concordance tables between national classifications and the classification used in the WIOD. The product classification corresponds to the classification used by Eurostat in its publications of SUTs for the EU27 countries. The choice for a 35 industry classification was based on the level of industry detail in the EU KLEMS database (O'Mahony and Timmer, 2009), which is the most important source of data for WIOD's socio-economic satellite accounts. Compared to that database, the textiles industry was further disaggregated to separate wearing apparel and footwear production which are both important industries in developing countries. We also provide some more detail for the transportation services industries as these are important for margin estimation and various environmental applications.

Price concepts: the price concept for gross outputs (basic prices) and intermediate inputs (purchasers' prices) has been harmonized across countries. Several countries, however, used a different price concept, including China and Japan (producers' prices), India and the USA (factor costs). Later in this section, we will discuss how the use tables in purchasers' prices were transformed into basic prices.

Harmonization involved aggregation and sometimes disaggregation based on additional detailed data. While for most European countries (due to a high level of harmonization of statistics in the EU), this was relatively simple, tables for non-EU countries proved more difficult. While aggregation of products or industries in a SUT is straightforward, disaggregation is not. To disaggregate an industry, first additional data from National Accounts was collected to breakdown value added and gross output by the sub-industry. To disaggregate an industry in a supply table, we assumed common product sales shares of the sub-industries.¹¹ To disaggregate an industry in a use table, we assumed common intermediate input coefficients for the sub-industries. Disaggregating products is more difficult because additional data by product are often not available (although in some cases a rough estimate could be made based on more detailed industry information). Disaggregating products in the supply table was based on common industry-production shares and, similarly, for the use table, we assumed common use shares.

National SUTs were also checked for consistency and adjusted to common concepts (e.g. regarding the treatment of financial intermediation services indirectly measured (FISIM) and purchases of residents abroad). In some cases, total supply and total use did not match at the product level, and differences were distributed across the final expenditure categories in order to balance supply and use. Undisclosed cells due to confidentiality concerns were imputed based on additional information and any negative entries in the intermediate block were reallocated. In particular, older SUTs do not have a

¹¹ Our philosophy of relying completely on official and publicly available data prevents us from disaggregating industries to lower levels. Lenzen (2011) provides evidence from Monte Carlo simulations that disaggregation – even on the basis of very incomplete data – is likely to produce more accurate multiplier values. Users of the WIOD database are encouraged to perform such disaggregation themselves, but the published database sticks to official numbers.

row allocation for FISIM in which case shares in financial services use or in value added were used.

Other aspects that were taken into account in the harmonization process are processing trade, re-exports, and transit trade. According to the System of National Accounts (SNA) 1993 (ISWGNA, 1994), following the Balance of Payments (BoP) manual, exports of goods and services consist of sales of goods and services from residents to non-residents, while imports consist of purchases of goods and services by residents from non-residents. This is the change-of-ownership principle. Goods that are in transit through a country are thus not to be included in export and import statistics, because they did not change ownership. Goods that are imported and exported again, without substantial change but that did change ownership (so-called re-exports), should be included.

The SNA 1993 recommends one exception to this change-in-ownership principle, namely for goods that are sent abroad for processing (without a change in ownership) and later re-imported (re-imports). These should be recorded gross by the processing economy as well as by the economy that sent the goods for processing, if the processing involves a substantial physical change in the goods (ISWGNA, 1994, p. 665). Unfortunately, national statistics offices have applied this recommendation in various ways. From our analytical perspective, any activity that generates value added should be included in the tables. Imports for processing should, therefore, always be recorded under intermediate consumption by the processing country, to reflect the underlying technology of the processing industry. This is also what we would like to see in our SUTs and IOTs. In practice, however, countries differ considerably in the application of this principle due to increasing reporting problems of processing firms. This has led to the new SNA 2008 (ISWGNA, 2009) recommendation to only record the processing fee as output and export of a service, and not the flow of intermediate imported goods (ISWGNA, 2009, p. 279). In the last decade, countries appear to differ widely in the treatment of processing trade and the actual treatment is often not well documented. For example, in the 2007 Chinese IOT, parts of imports for processing are excluded both in intermediate inputs and imports. On the other hand, many – but not all - European countries follow the SNA 1993 recommendation and record imports for processing as imports for intermediate consumption (e.g. Germany seems to be an exception).

This problem of recording imports is mirrored in the recording of exports and, thus, also statistical treatments for re-exports vary across countries. The US IOTs, for example, exclude re-exports and the associated imports, while most EU countries generally stick to the convention to include these.

In the WIOD project, we followed the SNA 1993 convention such that the output and the intermediate use of an industry best reflect the underlying production technologies. This means that, for example, a wearing apparel firm sewing shirts on the basis of imported parts should be represented by an intermediate flow of cloth and an output flow of shirts, instead of by the processing fee as output only (and no intermediate inputs). This holds, no matter which firm actually 'owns' the cloth parts and shirts in a juridical sense. We, therefore, add back re-exports, imports for processing purposes and exports of the processing industry into the original SUTs when this is needed and possible, notably for the USA and China. However, further research into this issue is warranted.

At this point, we have a published SUT for a given year that has been harmonized. Because we want to benchmark our data on National Accounts data, this harmonized SUT will be adapted (Section 3.2).

FIGURE 1. The national SUT.

	Products	Industries	Final use	Inventories	Exports	Total
Products		U	Y	s	е	q
Industries	\mathbf{V}_{b}					x _b
Imports	m ′					m
Value added		\mathbf{w}_b'				w _b
Margins	\mathbf{t}_m'					t_m
Net taxes	\mathbf{t}'_n					t_n
Total	q ′	\mathbf{x}_b'	y′	S	е	

where:

U is the 59×35 use table in purchasers' prices;

Y is the 59×4 matrix of final use in purchasers' prices (with four categories: consumption expenditures by households; consumption expenditures by non-profit organizations serving households, NPISH; consumption expenditures by the government; and gross fixed capital formation);

s is the 59×1 vector with changes in inventories by product and its total is given by s;

e is the 59×1 vector with exports (fob) and its total is given by *e*;

q is the 59×1 vector with total use in purchasers' prices;

 \mathbf{V}_b is the 35×59 supply table in basic prices (indicated by subscript *b*);

 \mathbf{x}_b is the 35×1 vector of gross output in basic prices by industry;

 \mathbf{m}' is the 1×59 vector of imports (cif) and its total is given by *m*;

 \mathbf{w}'_b is the 1×35 vector of value added by industry and its total is given by w_b ;

 \mathbf{t}'_m is the 1×35 vector of (i.e. trade and transport) margins, its total is given by t_m ;

 \mathbf{t}'_n is the 1×35 vector of net taxes (i.e. taxes minus subsidies), its total is given by t_n ;

 \mathbf{y}' is the 1×4 vector of final use totals;

 \mathbf{q}' is the 1×59 vector of total supply in purchasers' prices.

3.2. Benchmarking Harmonized National SUTs to the National Accounts

We start by describing the procedure for the years for which a full SUT is available, which is given in Figure 1.

Note that the use table is in purchasers' prices and the supply table in basic prices, and that the use table and final use matrix include imported products (which will be specified separately in the next section). Also note that $\iota' \mathbf{V}_b + \mathbf{m}'$ (where ι is used to indicate the summation vector with ones of appropriate length) gives the total supply in basic prices.

The national SUT in Figure 1 will be re-estimated on the basis of the following data that come from the National Accounts: total exports (\bar{e}) , total imports (\bar{m}) , gross output by the industry at basic prices $(\bar{\mathbf{x}}_b)$; value added by industry $(\bar{\mathbf{w}}_b)$; total final use at purchasers' prices $(\bar{\mathbf{y}}')$, total changes in inventories (\bar{s}) , total margins (\bar{t}_m) , and total net taxes (\bar{t}_n) . An over bar is used to indicate that the information was obtained from the National Accounts.

For benchmark years, the import and export vectors (i.e. $\hat{\mathbf{m}}$ and $\hat{\mathbf{e}}$) are estimated as follows:

$$\hat{m}_i = m_i \left(\frac{\bar{m}}{m}\right) \quad \text{and} \quad \hat{e}_i = e_i \left(\frac{\bar{e}}{e}\right).$$
 (1)

Supply tables are mostly in basic prices and often have additional information on margins and net taxes by product. The use table is typically in purchasers' prices and hence needs to be transformed into a basic price table in order to arrive at a common price concept. The difference between the two tables is given in the so-called valuation matrices (Eurostat, 2008, Chapter 6). These matrices are typically not available from public data sources and, therefore, need to be estimated. In the WIOD, we distinguish between margins (including all automotive trade, wholesale trade, retail trade and transport margins) and net taxes on products (taxes minus subsidies). The net tax rates by product are obtained from the supply tables and are retained to the extent possible. Tax rates are important elements in modelling applications that predict, e.g. the effects of changes in public tax systems. The margins are derived residually in two steps as follows.

First, the combination of margins plus net taxes is adapted to the National Accounts total. That is,

$$\hat{t}_{i} = \frac{(t_{i}^{m} + t_{i}^{n})(\bar{t}_{m} + \bar{t}_{n})}{t_{m} + t_{n}},$$
(2)

where t_i^m indicates the trade and transport margins for product *i* and t_i^n indicates the net taxes for product *i*. For the estimation of the net taxes, we use net tax rates as derived from the SUT, i.e. $\tau_i = t_i^n/q_i$. For products that have no margins (these are mainly services), we have (as a first estimate) $\tilde{t}_i^n = \hat{t}_i$. For products that do have margins, we have $\tilde{t}_i^n = \tau_i q_i$. Because the sum of these estimates of the net taxes does not sum to \bar{t}_n , they need to be normalized in the second step.

For products that have no margins, we have $\hat{t}_i^n = \tilde{t}_i^n = \hat{t}_i$. Note that these net taxes are not affected by the normalization, because they cannot differ from \hat{t}_i as the margins are zero. For products that do have margins, we have

$$\hat{t}_i^n = \tilde{t}_i^n \frac{\bar{t}_n - \sum_{i \in \{\text{no margin}\}} \tilde{t}_i^n}{\sum_{i \in \{\text{margin}\}} \tilde{t}_i^n},\tag{3}$$

where {no margin} is the set of products that have no margin and {margin} is the set of products that do have margins. The margins are then obtained as $\hat{t}_i^m = \hat{t}_i - \hat{t}_i^n$.

In order to derive IOTs in the end, both the supply and the use table should adopt the same price concept. Using the estimates of the margins and net taxes, the data in purchasers' prices in the use table are transformed into basic prices as follows. First, the rates of the margins and taxes by product are calculated, assuming that they do not apply to exports. This is a reasonable assumption as the bulk of the margins and taxes are applied when the product is sold to the final domestic consumer (retailing margins are much higher than other margins and taxes for most products) and not when the product is exported. For product *i* we then have

$$\hat{\tau}_{i}^{n} = \hat{t}_{i}^{n} / (\hat{q}_{i} - \hat{e}_{i}) \text{ and } \hat{\tau}_{i}^{m} = \hat{t}_{i}^{m} / (\hat{q}_{i} - \hat{e}_{i}).$$
 (4)

Next, this rate is uniformly applied to the *i*th row of the intermediate use table in purchasers' prices (**U**), to the *i*th row of the final use matrix in purchasers' prices (**Y**), and to s_i . This provides the valuation matrix for net taxes and for margins. Subtracting the valuation matrices from **U**, **Y** and **s** yields initial estimates for the use table in basic prices ($\hat{\mathbf{U}}_b^0$) and the final use matrix in basic prices ($\hat{\mathbf{Y}}_b^0$).

	Products	Industries	Final use	Inventories	Exports	Total
Products		$\widehat{\mathbf{U}}_{b}^{0}$	$\widehat{\mathbf{Y}}_{b}^{0}$	ŝ	ê	•••
Industries	\mathbf{V}_{b}					$\overline{\mathbf{x}}_{b}$
Net taxes		$\hat{\mathbf{t}}'_{\text{ind}}$	$\hat{\mathbf{t}}'_{\mathrm{fin}}$	$t_{\rm inv}$		\overline{t}_n
Imports	$\widehat{\mathbf{m}}'$					\overline{m}
Value added		$\overline{\mathbf{w}}_b'$				\overline{W}_{b}
Total		$\overline{\mathbf{x}}_b'$	$\overline{\mathbf{y}}'$	5	ē	

FIGURE 2. The national SUT, with estimates on the basis of National Accounts.

Unfortunately, data on changes in inventories by product are typically not collected by National Statistical Institutes on an annual basis and are occasionally measured as a residual in the attempt to balance supply and use at the product level. Given this uncertainly in the official estimates, we decided to adopt a procedure that would be robust, such that the estimation of the other elements in the SUTs would not be greatly affected. Our approach is to estimate them based on total changes in inventories as can be found in the National Accounts. We thus have

$$\hat{s}_i = s_i + \frac{|s_i|}{\sum_i |s_i|(\bar{s} - s)}.$$
 (5)

Note that the difference between the National Accounts total (i.e. \bar{s}) and the total in the SUT (i.e. *s*) is distributed according to absolute shares for products, and not proportionally. This is because proportionality would lead to sizable swings over time due to the fact that changes in inventories can be both positive and negative.¹²

Using the estimates obtained so far yields the situation as depicted in Figure 2. Note that the column sums of the valuation matrix for net taxes appear as a separate row (i.e. \hat{t}_{ind} , \hat{t}_{find}) and t_{inv}). It should be emphasized that the table in Figure 2 is not consistent in the sense that columns do not sum to the given column totals, for example. Figure 2 just describes the available information at this stage. The (blocks of) cells in Figure 2 that are coloured grey, contain data that are obtained directly from the National Accounts (using an over bar) or contain estimates that are benchmarked to the National Accounts (using hats). What remains is to estimate the matrices $\hat{\mathbf{U}}_b$, $\hat{\mathbf{Y}}_b$ and $\hat{\mathbf{V}}_b$ that make the table in Figure 2 consistent. This is done with the so-called SUT-RAS method developed by Temurshoev and Timmer (2011). This method is akin to the well-known bi-proportional updating method for IOTs known as the RAS-technique.¹³ This technique has been adapted for updating SUTs and has been shown to outperform other methods for the generation of (time-series of) SUTs. The SUT–RAS method requires initial estimates (which are given by $\hat{\mathbf{U}}_{b}^{0}$, $\hat{\mathbf{Y}}_{b}^{0}$ and \mathbf{V}_{b} in the uncoloured blocks) which are iteratively adapted, where the information in the grey cells is taken as given and fixed. Note that SUT-RAS does not require exogenous information for the total supply/use in basic prices. The SUT-RAS procedure results in estimates for the

¹² It should be stressed that changes in inventories may be very large. For example, the total changes in inventories amounted for Indonesia to 3.4% of GDP in 1997. Due to the economic crisis, many firms decided to eat into their reserves so that the total changes in inventories were no less than -8.7% of GDP in 1998 (and the differences between the 2 years are much larger at the level of single industries).

¹³ For an overview of earlier contributions to this literature, see Lahr and de Mesnard (2004). See, for example, Lenzen et al. (2007, 2009) and Minguez et al. (2009), for recent developments. Jackson and Murray (2004) and Temurshoev et al. (2011) tested the accuracy of several RAS-like methods against other updating procedures.

supply table $(\hat{\mathbf{V}}_b)$, the use table $(\hat{\mathbf{U}}_b)$ and the final use matrix $(\hat{\mathbf{Y}}_b)$, all in basic prices. The total supply/use in basic prices then follows from summation, which yields $\hat{\mathbf{q}}_b$.

3.3. Building a time series of national SUTs

In this subsection, we will describe how the national SUTs are estimated for any intermediate year t, given that published SUTs are available for years b_1 and b_2 (with $b_1 < t < b_2$). For each year t, all information in the final column and row of Figure 2 with an over bar is available from the National Accounts. Like it was the case in the previous subsection, we need to estimate imports/exports, changes in inventories, margins and net taxes at the product level before SUT–RAS is applied.

For exports in years b_1 and b_2 , Equation 1 is used, which yields $\hat{e}_i^{(b_1)}$ and $\hat{e}_i^{(b_2)}$ for the exports based on information from the published national SUTs. For intermediate years, the export data by product need to be interpolated. This is based on ITS. Interpolation is done using the annual growth rates of ITS at the product level. To accommodate the annual fluctuations but at the same time retain the levels in b_1 and b_2 , we employ a procedure that uses the movement of the ITS data with a correction for differences in average annual growth rates. If we denote the exports in year t of product i in the ITS by $ew_i^{(t)}$ we have

$$\hat{e}_{i}^{(t)} = \hat{e}_{i}^{(t-1)} \cdot \frac{ew_{i}^{(t)}}{ew_{i}^{(t-1)}} \cdot \sqrt[b_{2}-b_{1}]} \frac{\hat{e}_{i}^{(b_{2})}/\hat{e}_{i}^{(b_{1})}}{ew_{i}^{(b_{2})}/ew_{i}^{(b_{1})}}.$$
(6)

Next, these estimates need to be normalized such that they add to the total exports ($\bar{e}^{(t)}$) from the National Accounts. That is,

$$\hat{\hat{e}}_{i}^{(t)} = \bar{e}^{(t)} \frac{\hat{e}_{i}^{(t)}}{\sum_{i} \hat{e}_{i}^{(t)}}.$$
(7)

Imports at the product level are estimated in the same way.

For the margins and net taxes, we can follow the same steps as outlined above, including Equations 2–4. For this, we can take two starting points, b_1 or b_2 . This also leads to two outcomes for the initial estimate (for year *t*) of the use table in basic prices, for example. Let us denote them by $\hat{\mathbf{U}}_{b}^{(b_1-\text{based})}$ and $\hat{\mathbf{U}}_{b}^{(b_2-\text{based})}$. The SUT–RAS is initialized with an average as follows

$$\hat{\mathbf{U}}_{b}^{(t)} = \frac{(b_{2}-t)\hat{\mathbf{U}}_{b}^{(b_{1}-\text{based})} + (t-b_{1})\hat{\mathbf{U}}_{b}^{(b_{2}-\text{based})}}{b_{2}-b_{1}}.$$
(8)

The same applies to the other two matrices that are required to initialize SUT-RAS. That is,

$$\hat{\mathbf{Y}}_{b}^{(t)} = \frac{(b_2 - t)\hat{\mathbf{Y}}_{b}^{(b_1 - \text{based})} + (t - b_1)\hat{\mathbf{Y}}_{b}^{(b_2 - \text{based})}}{b_2 - b_1}$$

and

$$\hat{\mathbf{V}}_{b}^{(t)} = \frac{(b_2 - t)\mathbf{V}_{b}^{(b_1)} + (t - b_1)\mathbf{V}_{b}^{(b_2)}}{b_2 - b_1},$$

where $\mathbf{V}_{h}^{(b_{1})}$ indicates the supply table in year b_{1} .

For the changes in inventories, the procedure is similar to Equation 2. From the National Accounts, we have information on totals, i.e. $\bar{s}^{(t)}$. From the published SUTs, we used changes in inventories at the product level, i.e. $s_i^{(b_1)}$ and $s_i^{(b_2)}$, and their totals $s^{(b_1)}$ and $s^{(b_2)}$. Proceeding in a forward way gives us

$$\hat{s}_{i}^{(t)} = s_{i}^{(b_{1})} + \frac{\left|s_{i}^{(b_{1})}\right|}{\Sigma_{i}\left|s_{i}^{(b_{1})}\right|} (\bar{s}^{(t)} - s^{(b_{1})}).$$
(9a)

In the same fashion, proceeding in a backward way yields

$$\hat{s}_{i}^{(t)} = s_{i}^{(b_{2})} + \frac{\left|s_{i}^{(b_{2})}\right|}{\sum_{i}\left|s_{i}^{(b_{2})}\right|} (\bar{s}^{(t)} - s^{(b_{2})}).$$
(9b)

We have used the simple (arithmetic) average of the two outcomes as our estimate for the changes in inventories at the product level in year t.

The estimates so far, provide the same situation as depicted by Figure 2. The final step is to apply the SUT–RAS method again. This yields the supply table, the use table and the final use matrix, all for year t and all in basic prices. The total supply/use follows again from summation. Extrapolation for years before the first official SUT ($t < b_1$) and after the most recent one ($t > b_2$) are produced from initial estimates based on a single table only.

4. CONSTRUCTION OF THE INTERNATIONAL SUTS

The starting point for this stage is a national SUT with all the information in basic prices. As mentioned earlier, the intermediate use and the final use matrices include imported goods and services. In this section, it will be described how the imports are separated from the domestic deliveries. After that the imports of goods and services will be further broken down according to the country of origin (including the RoW as a single 'country'). We have termed the resulting tables as 'international SUTs'. For this procedure, we will use information from the bilateral trade database that was constructed within the WIOD project (see Foster et al., 2012, for details). Section 4.1 will describe the bilateral trade database and Section 4.2 will discuss the split of the national SUT so as to obtain an international SUT.

4.1. The Bilateral Trade Database

The bilateral trade database covers trade in goods and trade in services. The raw data for the trade in goods were taken from the UN Comtrade database (http://comtrade.un.org/db/default.aspx) and downloaded at the six-digit level in the Harmonized System (HS). The trade database contains the 40 WIOD countries over the period 1995–2010 as reporter countries and all other countries as partner countries. The data for

most countries are in HS 1996 for the years 1996–2010 and in HS 1992 for 1995. In some cases, trade data are missing and some gaps have been filled using data from other sources or have been gathered from the respective National Statistical Institutes (NSIs) on request.

In a next step, confidentiality issues had to be tackled. Such confidentiality issues arise when NSIs decide not to report either which amount of a certain product they have been trading (i.e. product-related confidentiality) or with whom they have been trading (i.e. partner-related confidentiality). This is typically the case when there is a near-monopoly situation for a specific product in a country. To allow for this, NSIs report this trade in an artificial partner country indicated as 'Special Categories' or 'Areas, nes'. Such trade flows are included in the information on total imports and exports in the national SUT and need to be attributed. Because an alternative is lacking, trade with partners 'Special Categories' and 'Areas, nes' has been distributed (proportionally to the reported trade values) among the other partner countries according to the difference between total reported trade and the sum of trade for the HS six-digit codes.

There were also some problems specific to particular countries, which had to be handled separately. For example, two particularly thorny issues related to China's trade data. First, data for trade with Taiwan do not exist in the UN Comtrade. Second, the special administrative regions (SARs) of China, i.e. Hong Kong and Macao, were included separately in the trade data. We have handled China and the SARs as one economic entity as is also done in the national SUTs for China. Therefore, trade with (or of) China, therefore, includes the trade with (or of) mainland China, Hong Kong and Macao, which is consistent with the trade reported in the Chinese national SUTs. Flows between China and the SARs have been netted out as well as re-imports and re-exports. We have used the reporter countries' trade with 'Other Asia, nes' to identify trade with Taiwan as a partner, following common practice. Trade data for Taiwan as a reporter country were obtained through the OECD.

Starting from HS six-digit which provides information on bilateral flows of goods of about 5,000 products, the individual flows were merged with a correspondence of the HS six-digit products to use categories distinguishing 'intermediate consumption', 'final consumption', and 'capital goods'. This correspondence was constructed from the Broad Economic Categories (BEC revision 3) classification as provided by UN and the correspondence between these detailed end-use categories into broader groups as applied by OECD. For about 700 products, the correspondence to a particular use category was, however, revised by reclassifying products to the above-mentioned three categories. The classifications do not deal with the problem that one particular good might qualify for more than one use category. For example, cars are both used as final consumption and as investment, and motor spirits as intermediate inputs by firms or as final use by consumers. Therefore, for such products at the HS six-digit level, weights have been assigned (applying simple rules like 1/3 - 2/3 or 1/2 - 1/2) allowing for a classification into our three end-use categories.¹⁴ Furthermore, the HS six-digit data were merged with a correspondence to NACE revision 1 at the two-digit level as made available by the Eurostat corresponding to the CPA classification in the national SUTs.

¹⁴ This is still a rather rough procedure. In principle, reporter–product specific (if not reporter–partner–product specific) weights would have to be assigned (e.g. for imports/exports of shirts and parts of shirts which are classified in one HS six-digit category).

The procedure of estimating the bilateral import flows in the national SUTs is as follows.¹⁵ The share of the imports of (each of approximately 5,000) product *j* that has country *h* as origin and that goes to each of the three end-use categories is given by , $\alpha_{jh}^{\text{interm}}$, $\alpha_{jh}^{\text{finalcons}}$ and $\alpha_{jh}^{\text{gfcf}}$ and are derived from the ITS. We will use m_j^{ITS} to indicate the total imports of product *j* in the ITS, because they differ from the imports as given in the national SUTs (which will be indicated later by m_i^{SUT}). This implies

$$m_{jh}^{\text{interm}} = \alpha_{jh}^{\text{interm}} m_j^{\text{ITS}}; m_{jh}^{\text{finalcons}} = \alpha_{jh}^{\text{finalcons}} m_j^{\text{ITS}}; m_{jh}^{\text{gfcf}} = \alpha_{jh}^{\text{gfcf}} m_j^{\text{ITS}}$$

with $\Sigma_h(\alpha_{jh}^{\text{interm}} + \alpha_{jh}^{\text{finalcons}} + \alpha_{jh}^{\text{gfcf}}) = 1$. Next the products are aggregated into the 59 product groups of the WIOD classification. Each of the WIOD-products i (= 1, ..., 59) consists of $1, ..., n_i$ HS six-products. We thus have for i = 1, ..., 59;

$$m_{ih}^{\text{interm}} = \sum_{j=1}^{n_i} m_{jh}^{\text{interm}}; \ m_{ih}^{\text{finalcons}} = \sum_{j=1}^{n_i} m_{jh}^{\text{finalcons}}; \ m_{ih}^{\text{gfcf}} = \sum_{j=1}^{n_i} m_{jh}^{\text{gfcf}}$$

For the total imports of each WIOD-product, we have $m_i^{\text{ITS}} = \sum_{j=1}^{n_i} m_j^{\text{ITS}}$. At the level of WIOD-products, the share of the imports originating from a country *h* that goes to each of the end-use categories is then given by

$$\alpha_{ih}^{\text{interm}} = m_{ih}^{\text{interm}} / m_i^{\text{ITS}}; \ \alpha_{ih}^{\text{finalcons}} = m_{ih}^{\text{finalcons}} / m_i^{\text{ITS}}; \ \alpha_{ih}^{\text{gfcf}} = m_{ih}^{\text{gfcf}} / m_i^{\text{ITS}}.$$
(10)

For the construction of the bilateral data on services flows across countries, it is important to emphasize that services have unique characteristics that greatly affect their tradability and the ability to observe them by current statistical methods. The two most obvious characteristics include intangibility and in many cases of non-storability. In addition, they often require customers having to participate in the production process (Francois and Hoekman, 2010). To capture these aspects, the WTO defines in its General Agreement on Trade in Services (GATS), services trade to span the following four modes of supply. Mode 1 – Cross-border: services supplied from the territory of one country into the territory of another, Mode 2 – Consumption abroad: services supplied in the territory of a nation to the consumers of another, Mode 3 – Commercial presence: services supplied through any type of business or professional establishment of one country in the territory of another (i.e. FDI), and Mode 4 -Presence of natural persons: services supplied by nationals of a country in the territory of another. In the data set collected for the WIOD, only data on cross-border services trade in the GATS mode 1 has been used. The WIOTs are constructed on a territorial basis meaning that they include all activities that take place on the territory of the country, either by residents or non-residents, so mode 3 and 4 are not considered as part of imports and exports. Mode 2 activities are already covered by the items 'purchases of non-residents on domestic territory' and 'foreign purchases of residents' in the national SUTs and are not split further by the country of supply.

¹⁵ Throughout the rest of this subsection, we will use the index j for the 5000 products at the HS six-digit level and the index i for the 59 products in the WIOD SUTs.

To create a data set covering bilateral services flows data from the UN, Eurostat, and OECD have been collected and used.¹⁶ These sources provide bilateral services trade flows with a dual breakdown, i.e. by partners and by BoP codes. The most comprehensive coverage of reporting countries among the three sources is UN, which provides data for 190 reporter countries. Eurostat and OECD provide data for a limited number of reporters: Eurostat covers 27 EU members plus Croatia, Iceland, Japan, Norway, Turkey, Switzerland, and USA; while OECD covers 28 countries (all the OECD members apart from Chile, Iceland, Israel, Slovenia, and Switzerland). The time coverage is the largest in EUROSTAT, which reports data starting from 1995. Since the quality of existing services data is rather poor as compared with merchandize trade statistics, the data had to be adjusted extensively to assure their consistency.

Working with multiple sources made it somewhat easier to identify problems in the data linked to simple errors. For example, there were clear cases where one source reports flows three times larger than the other two sources. Based on the examination of the data, comparisons across sources have been made to filter errors of this type, as well as other problems apparently linked to identifiable data entry errors. It became clear from this examination that the UN sources have (in many cases) identified and cleaned up errors that remain in the EUROSTAT and OECD series. For this reason, the UN data have become the preferred data source.

The resulting data set contains data on bilateral services trade flows for the 40 WIOD countries as reporters and partners plus the RoW. Additionally, for each of the countries data on trade with the world is available. It includes more than 20 economic activities according to the BoP classification. However, not all sectors have the same coverage in terms of time and trading countries. In general, the higher the level of disaggregation, the fewer observations are available. As a word of caution, it should be added that the quality of trade data in services is still far away from being comparable to trade data for merchandize goods. Due to the importance for collecting tariff revenues, trade data for goods have been collected with quite high quality and accuracy. Due to intangibility and nonstorability of services, at-the-border-duties cannot be applied to services, which have resulted in compilation practices with considerably less accuracy. There is ample space for further improvements in the measurement of services trade. The WIOD database for trade in services should be seen in this light as the best currently available approximation to a comprehensive picture of global trade flows in Mode 1 services.

4.2. Estimation of the Use of Imports in the SUTs

This subsection describes how the international SUTs have been obtained from combining national SUTs with information from the bilateral trade database. First, all data had to be converted into current US-\$. The exchange rates (year averages) applied were collected from the International Financial Statistics database of the International Monetary Fund (IMF). This resulted in a set of 40 national SUTs for each year over the period 1995–2009 expressed in (millions of) current US-\$. The decision to convert national currencies into US-\$ (rather than, for example, euros) was motivated by the fact that trade statistics are also

¹⁶ See http://unstats.un.org/unsd/servicetrade/default.aspx, http://epp.eurostat.ec.europa.eu/portal/page/portal/ balance_of_payments/data/database, http://www.oecd.org/std/internationaltradeandbalanceofpaymentsstatistics/ oecdstatisticsoninternationaltradeinservicesdetailedtablesbypartnercountry2004-20072009edition.htm

reported in US-\$ terms, but any other numéraire currency could have been applied and the choice would have had no consequence for the construction process.

The supply table provides information for the imports (in cost, insurance and freight; cif prices) of each of 59 products (given by the row vector $\widehat{\mathbf{m}}'$ in Figure 2). Let us denote the imports of product *i* by m_i^{SUT} (in contrast to the imports m_i^{ITS} from ITS data). The use table gives the use of product *i* by industry *j* as u_{ij} , which is split into the use of product *i* that is domestically produced and the part that is imported. For this, we have used the shares given in Equation 10. That is, for i = 1, ..., 59 and j = 1, ..., 35,

$$u_{ij}^{\text{IMP}} = \frac{u_{ij}}{\sum_{k=1}^{35} u_{ik}} \left(\sum_{h} \alpha_{ih}^{\text{interm}} \right) m_i^{\text{SUT}} \quad \text{and} \quad u_{ij}^{\text{DOM}} = u_{ij} - u_{ij}^{\text{IMP}}, \tag{11}$$

where the index h indicates the country of origin.

The same procedure is applied to the final consumption categories, with final consumption expenditures by households, by non-profit organizations serving households (NPISH), and by the government (i.e. the first three columns of the matrix **Y** in Figure 1). That is, for i = 1, ..., 59 and j = 1, ..., 3,

$$y_{ij}^{\text{IMP}} = \frac{y_{ij}}{\sum_{k=1}^{3} y_{ik}} \left(\sum_{h} \alpha_{ih}^{\text{finalcons}} \right) m_i^{\text{SUT}} \text{ and } y_{ij}^{\text{DOM}} = y_{ij} - y_{ij}^{\text{IMP}}.$$
(12)

For gross fixed capital formation (which is the fourth column of the matrix **Y** in Figure 1 and is, therefore, indicated by y_{ij}^{IMP} with j = 4), we have

$$y_{i4}^{\text{IMP}} = \left(\sum_{h} \alpha_{ih}^{\text{gfcf}}\right) m_i^{\text{SUT}} \text{ and } y_{i4}^{\text{DOM}} = y_{i4} - y_{i4}^{\text{IMP}}.$$
 (13)

Note that we have made some assumptions here. First, it was assumed that changes in inventories are only related to domestic production. Similarly, we assumed that exports can only be made out of domestic production.

It should be emphasized that it occasionally has happened that the imports as given in the national supply tables (i.e. m_i^{SUT} , which equals \hat{m}_i in Figure 2) were larger than the total domestic use $\sum_j u_{ij} + \sum_k y_{ik}$. By definition, this also means that exports are larger than domestic production. In those cases, we have defined 're-exports' as $m_i^{\text{SUT}} - (\sum_j u_{ij} + \sum_k y_{ik})$ and these have been subtracted from the total imports. Consequently, in cases where imports were larger than total domestic use, we have taken $u_{ij}^{\text{IMP}} = u_{ij}$ and $u_{ij}^{\text{DOM}} = 0$ for all j = 1, ..., 35, and $y_{ik}^{\text{IMP}} = y_{ik}$ and $y_{ik}^{\text{DOM}} = 0$ for all k = 1, ..., 4. Total imports change to m_i^{SUT} minus re-exports, which equals $\sum_j u_{ij} + \sum_k y_{ik}$. Note that re-exports thus defined are assumed not to be part of the domestic production process.

The shares of re-exports in total exports (i.e. exports plus re-exports as defined above) vary largely across countries, going up to 18% in the Netherlands and 14% in Cyprus (averaged over the years). Also other small economies (e.g. Lithuania, Estonia, and Luxembourg) have large shares (between 5% and 10%). In more than 30 countries, however, the shares are smaller than 5% and in about half of the countries they are lower than 2%, while they are even zero for Japan, China and Brazil. Overall, the shares tend to increase over time. With respect to products, the shares can be very high in some countries for particular products

(e.g. mining). In the manufacturing sectors, large shares of re-exports are found for tobacco products (CPA 16) in the case of Slovakia and Slovenia, for leather and leather products (CPA 19) in Belgium, Denmark, Luxembourg and the Netherlands, for coke and refined petroleum products (CPA 23) in Cyprus and Slovenia, and for electronic products (CPA 30 to CPA 33) in a number of countries. The next step is to split the use of imported products according to the country of origin. Using the shares from Equation 10, we have

$$u_{ij}^{h} = \frac{\alpha_{ih}^{\text{interm}}}{\sum_{h} \alpha_{ih}^{\text{interm}}} u_{ij}^{\text{IMP}}.$$
 (14)

For any of 40 target countries in the WIOD, the index h covers each of the remaining 39 WIOD countries and the RoW, which is obtained from summing over all countries in the trade database that are non-WIOD countries. A similar procedure was also used for the final use and gross fixed capital formation,

$$y_{ij}^{h} = \frac{\alpha_{ih}^{\text{finalcons}}}{\sum_{h} \alpha_{ih}^{\text{finalcons}}} y_{ij}^{\text{IMP}} \quad \text{and} \quad y_{i4}^{h} = \frac{\alpha_{ih}^{\text{gfcf}}}{\sum_{h} \alpha_{ih}^{\text{gfcf}}} y_{i4}^{\text{IMP}}.$$
 (15)

A major advantage of this procedure is that the imports of goods do not rely on the standard proportionality assumption that is popular in the literature and applied in various other databases, see, for example, Peters et al. (2011) and Johnson and Noguera (2012). In those cases, a common import proportion is used for all cells in a use row, irrespective of the user. This common proportion is simply calculated as the share of imports in total supply of a product. We find that import proportions differ widely across use categories and importantly, within each use category they differ also by country of origin.¹⁷ Our detailed bilateral approach ensures that this type of information is reflected in the international SUTs and consequently the WIOT. To highlight the differences between the two assumptions, an example is given in Table 1. Table 1 gives a hypothetical example of computers that are used by two industries (financial intermediation and transport equipment), by two categories of final consumption (households and government), for gross fixed capital formation, and for exports. The total uses (including imports) are given in row (1). Exports are sourced only from domestic production. Suppose that it is given that the imports from the USA are 60 and from the RoW 40. Total use minus exports amounts to 200 and therefore 30% of domestic use are imports from the USA and 20% from RoW. Applying the standard proportionality assumption means that the import proportion for the USA (30%) is the same for all uses (except exports, which can only be domestically sourced). Hence, row (2) is obtained by multiplying row (1) with 0.3, except for the exports. Similarly, row 3 uses an import proportion of 20% for RoW. The use of domestic goods is obtained as (1) - (2) - (3), which (for the standard proportionality assumption) is the same as multiplying row (1) with the proportion (i.e. 50%) of domestic use.

In the lower part of Table 1, i.e. rows (5)–(7), we have applied the following shares: $\alpha_{\text{US}}^{\text{interm}} = 0.30$; $\alpha_{\text{US}}^{\text{finalcons}} = 0.25$; $\alpha_{\text{US}}^{\text{gfcf}} = 0.05$; $\alpha_{\text{RoW}}^{\text{interm}} = 0.32$; $\alpha_{\text{RoW}}^{\text{finalcons}} = 0.05$; $\alpha_{\text{RoW}}^{\text{gfcf}} = 0.03$. The imports are 62 for intermediate use, which are proportionally distributed within this category, thus 31 to financial intermediation and 31 to transport equipment. The imports for final use are 30, 18 for household consumption and 12 for government consumption.

¹⁷ See Puzzello (2012) for more empirical evidence of the implausibility of the proportionality assumption.

	Inter	Intermediate		mption			
Computers	Fin Int	Trans Eq	Househ	Govern	Gfcf	Exports	Total
(1)	70	70	30	20	10	10	210
(2) USA = 30% (3) RoW = 20% (4) DOM = 50%	21 14 35	21 14 35	9 6 15	6 4 10	3 2 5	0 0 10	60 40 110
(5) USA (6) RoW (7) DOM	15 16 39	15 16 39	15 3 12	10 2 8	5 3 2	0 0 10	60 40 110

TABLE 1. An example to illustrate the effect of different assumptions for imports.

The imports for gross fixed capital formation amount to 8. The domestic use follows from Equations 11–13, and the split of the imports between the USA and the RoW from Equations 14 and 15.

In the real data, the German imports of CPA 34 (Motor vehicles, trailers, and semi-trailers) and CPA 35 (Other transport equipment) were at about the same level for intermediate use and final use in 1995. However, the import share of these products in 2008 had more than quadrupled for intermediate use whereas it only increased by 70% for final use. The 1995 German imports of these two products for intermediate use were about 50% of the total import use (i.e. for intermediate plus final use) and this share increased to 70% in 2008. In the Czech Republic, a target country for offshoring, this share was already 65% in 1995 and increased to more than 80% in 2008. So the application of a non-proportional approach is not only a theoretical refinement, but is also empirically important.

Derivation of import use proportions for services trade is more complicated than for goods. Unfortunately, the BoP data for services trade does not provide any indication on the use of the services imported. For a rough estimate, we have examined information from existing import IO tables from Eurostat which are available in the product by product dimension. These tables indicate which shares of services imports by CPA categories are used as intermediates and final goods. However, close inspection revealed huge differences across countries and over time for individual countries. Therefore, we used a simple average of these shares across countries and over time which has been applied to all years and countries. On average, the bulk of service imports (about 70%) is intermediate imports. A second remark on services trade is that a correspondence between BoP codes and CPA had to be made in order to derive the bilateral shares (the levels were given in the national SUTs). Only a very rough match is possible, given the limitations of detailed bilateral services trade data for particular BoP codes and the differences in classifications of BoP versus CPA codes.

A final remark is with respect to the international trade and transport margins. So far, imports are given in cif prices. In order to match them with export flows, which are expressed in 'free on board' (fob) prices, the imports are converted from cif prices into fob prices. This required the estimation of bilateral international trade margins by the product category. Using detailed trade data from the UN Comtrade database, the ratio of import unit values and export unit values has been estimated in a gravity model with distance, land-lockedness and dummy for the same continent as explanatory variables. This resulted in a set of bilateral

	Products	Industries	Final use	Inventories	Export	Total
Products, country A		\mathbf{U}^{AA}	Y ^{AA}	\mathbf{s}^{A}	\mathbf{e}^{A}	$\widetilde{\mathbf{q}}^{A}$
Imports (fob) from country B		\mathbf{U}^{BA}	\mathbf{Y}^{BA}			$\widetilde{\mathbf{m}}^{BA}$
Imports (fob) from RoW		\mathbf{U}^{RA}	\mathbf{Y}^{RA}			$\widetilde{\mathbf{m}}^{RA}$
Industries, country A	\mathbf{V}^{A}					\mathbf{x}^{A}
Imports (cif) by country A	$(\mathbf{m}^{BA} + \mathbf{m}^{RA})'$					m^A
Value added		$(\mathbf{w}^A)'$				w^A
International transport margins		$(\mathbf{tm}_{\mathbf{U}}^{A})'$	$(\mathbf{tm}_{\mathbf{Y}}^{A})'$			tm^A
Total	$(\mathbf{q}^A)'$	$(\mathbf{x}^{A})'$	$(\mathbf{y}^A)'$	S^A	e^{A}	
NT . TIBA TIRA TIBA TIRA NARA ~ BA 1~ B	4 11 1	0.1 .				

Note: \mathbf{U}^{BA} , \mathbf{U}^{RA} , \mathbf{Y}^{BA} , \mathbf{Y}^{RA} , $\mathbf{\tilde{m}}^{BA}$ and $\mathbf{\tilde{m}}^{RA}$ are all expressed in fob prices.

cif–fob margins for each CPA product.¹⁸ These have further been benchmarked by the trade balance of the WIOD countries in the margins sectors (transport and trade) as outlined in Streicher and Stehrer (2012). This results in estimated overall transport costs of 5-7% of imports on average.

The final outcome of these procedures are import use tables (distinguishing between countries of origin) in cif terms from which the bilateral trade and transport margins were subtracted to generate the use tables in fob terms. The imported use in fob terms can be aligned with the export vector of the origin country when constructing the world SUTs in the next section. Figure 3 sketches the international SUT for country A, assuming that there is one other country (i.e. B) in the database and that there is a RoW. Potential users can download the 1995–2009 time series of international SUTs of this type for all 40 countries included in the WIOD.

Note that $\tilde{\mathbf{q}}^A$ gives the total use of *domestically* produced products (which equals $\iota' \mathbf{V}^A$) and that \mathbf{q}^A gives total supply of products.

5. CONSTRUCTION OF THE WIOTS

This section describes how the international SUTs for each country discussed in the previous section have been transformed into symmetric WIOTs of the industry-by-industry type for the whole set of countries.

The international SUTs have a product-by-industry nature. Symmetric WIOTs could thus be of the product-by-product type, or the industry-by-industry type. Both types of tables have their advantages (see Chapter 5 in Miller and Blair, 2009, and references therein). Most analytical work links IOTs to additional data available at the industry level, like the data available in the socio-economic and environmental satellite accounts in the WIOD. Consideration of the potential use of the data led to the decision to construct WIOTs of the industry-by-industry type.¹⁹ To produce a WIOT of the industry-by-industry type from international SUTs, two steps were taken. First, the international SUTs have been merged into what we call a 'world SUT'. Next, the world SUT was transformed into a WIOT, by using the fixed product sales structure assumption.

¹⁸ See Streicher (2012) for further details.

¹⁹ Because the international SUTs are available in the WIOD, interested users could produce WIOTs of the product-by-product type themselves. The relative advantages of various methods to construct such tables have been extensively discussed in Eurostat (2008).

	Α	В	Α	В	Α	В		RoW	Total
	products	products	industries	industries	final use	final use	Invent	exports	
Products, A			U ^{AA}	\mathbf{U}^{AB}	Y ^{AA}	\mathbf{Y}^{AB}	s ^A	e ^{AR}	$\widetilde{\mathbf{q}}^{A}$
Products, B			\mathbf{U}^{BA}	\mathbf{U}^{BB}	\mathbf{Y}^{BA}	\mathbf{Y}^{BB}	\mathbf{s}^{B}	e^{BR}	$\widetilde{\mathbf{q}}^{B}$
Products, RoW	Ī		\mathbf{U}^{RA}	\mathbf{U}^{RB}	\mathbf{Y}^{RA}	\mathbf{Y}^{RB}			
Industries, A	\mathbf{V}^{A}								\mathbf{x}^{A}
Industries, B	Ī	\mathbf{V}^{B}							\mathbf{x}^{B}
Imports (cif)	$(\mathbf{m}^{BA})'^+$ $(\mathbf{m}^{RA})'$	$(\mathbf{m}^{AB})'^+$ $(\mathbf{m}^{RB})'$							
Value added			$(\mathbf{w}^4)'$	$(\mathbf{w}^B)'$					
International transport margins			$(\mathbf{tm}_{\mathbf{U}}^{A})'$	$(\mathbf{tm}_{\mathbf{U}}^{B})'$	$(\mathbf{tm}_{\mathbf{Y}}^{A})'$	$(\mathbf{tm}_{\mathbf{Y}}^{B})'$			tm
Total	$(\mathbf{q}^A)'$	$(\mathbf{q}^{\overline{B}})'$	$(\mathbf{x}^{A})'$	$(\mathbf{x}^{\overline{B}})'$					

5.1. Constructing World SUTs

In what follows, we continue the hypothetical 'two-country plus RoW' example from the previous section. Figure 3 gave the international SUT for country *A* and a similar SUT is available for country *B*. The world SUT in this case is obtained by combining the information and rearranging the original information, which yields Figure 4.

In Figure 4, the intermediate inputs parts of the use tables for country *A* are exactly the same as those given in Figure 3. Note, however, that the exports of intermediate products from country *A* to country *B* were obtained during the construction of the international SUT for country *B*. The same applies to the exports of final products from *A* to *B*. This is by way of construction: we assume that the exports from *A* to *B* are mirror flows of the imports of *B* from *A* which have been calculated as described above. Because exports are in fob terms and because exports are taken as the mirror flows of imports, it was necessary in Section 4.2 to transform imports in cif terms to imports in fob terms. Total exports of country *A* in the international SUT was given by e^A and exports to country *B* amount to $U^{AB}\iota + Y^{AB}\iota$. Exports to RoW can thus be derived as a residual (i.e. as $e^A - U^{AB}\iota - Y^{AB}\iota$).

There are downsides to this residual approach to determine exports to RoW. All measurement errors, aggregation biases, inconsistencies between National Accounts and ITS, and other problems that pertain to the trade flows among the WIOD countries tend to accumulate in the residual. In particular, it is possible that exports to ROW when measured residually become negative. This is undesirable and, therefore, we ruled out this possibility by creating an additional trade reporter 'RoW' in the previous step when trade shares were calculated, described in Section 4.1. For each product at the HS six-digit level, the world aggregate (i.e. the sum over all countries) of exports and of imports were downloaded from the UN Comtrade database. Subtracting the exports and imports by the WIOD reporters from this world aggregate provided the exports and imports of RoW. For trade in services, the available trade in services database in the WIOD has been used.

Using this information, a RAS procedure was applied to reconcile bilateral trade flows. This yielded a bilateral trade matrix and total imports and exports for each product for all WIOD countries and RoW. Moreover, the imports and exports totals are in line with the information from the national SUTs. It should be noted that this procedure adjusted the bilateral trade shares when compared to the shares from the original trade in goods and services data as described in Section 4.1. These adjusted trade data have then been used to construct the international SUTs following the steps reported in Section 4.2.

5.2. From World SUTs to WIOTs

The final part was the transformation of the world SUT into a symmetric WIOT. Note that the world SUT can be considered as a detailed (and partitioned) national SUT for the world economy. Hence, well-known transformation methods to obtain symmetric IOTs from a SUT system could be applied. Given the choice to derive WIOTs of the industry-by-industry type, two transformation methods could be considered (see Rueda-Cantuche and ten Raa, 2009, for an extensive discussion of alternative methods to derive industry-by-industry IOTs from SUTs). The difference between the two methods is in the treatment of secondary products. The supply table in the world SUT also contains a lot of positively valued cells that do not relate to the main products of an industry (since the number of products exceeds the number of industries in the WIOD, this phenomenon emerges by construction). Eurostat (2008, Chapter 11) provides a very accessible explanation of the main characteristics of both methods, which are commonly referred to as 'Model C' and 'Model D'.

Model C treats secondary products according to the *fixed industry sales structure* assumption. This assumption boils down to assuming that each industry has its own sales structure (for intermediate input deliveries to other industries, and to various final demand categories). This implies that it sells its secondary products to exactly the same industries and final users and in exactly the same proportions as its main products. In Model D, the *fixed product sales structure* assumption is the point of departure. In this case, each product has its specific sales structure. It does not matter in which industry the product shat have been produced as an industry's principal output or as a secondary product by another industry.

Practitioners have developed a clear preference for Model D. This preference is based on two major disadvantages of Model C. First, and most importantly, the fixed industry sales structure is considered to be much more implausible in an empirical sense than the fixed product sales structure assumption. Secondly, application of Model C can yield IOTs with negative entries in columns where only nonnegative values should appear (i.e. in the intermediate inputs block and in consumption), even if the original SUT does not. Some *ad hoc* methods have been proposed to solve these problems, but the two drawbacks of Model C taken together have been sufficient to make Model D the clearly dominant transformation method. In the WIOD, this transformation method has been applied as well.

The mathematical expressions that are involved in the transformation according to Model D can be found in Eurostat (2008, p. 349). That is, define the industry-by-product matrix $\mathbf{B}^{A} = \mathbf{V}^{A}(\hat{\mathbf{q}}^{A})^{-1}$ with its element b_{ij}^{A} indicating the share of domestic output of product *j* that is produced by industry *i*. A similar definition holds for \mathbf{B}^{B} . Figure 5 depicts the WIOT for the case of countries *A* and *B*, and a RoW. The industry-by-industry matrices of intermediate deliveries are then given by

$$\mathbf{Z}^{AA} = \mathbf{B}^{A}\mathbf{U}^{AA}, \quad \mathbf{Z}^{AB} = \mathbf{B}^{A}\mathbf{U}^{AB}, \quad \mathbf{Z}^{BA} = \mathbf{B}^{B}\mathbf{U}^{BA}, \text{ and } \mathbf{Z}^{BB} = \mathbf{B}^{B}\mathbf{U}^{BB}.$$

Similarly, we have for the final use matrices at the industry level

$$\mathbf{F}^{AA} = \mathbf{B}^{A}\mathbf{Y}^{AA}, \quad \mathbf{F}^{AB} = \mathbf{B}^{A}\mathbf{Y}^{AB}, \quad \mathbf{F}^{BA} = \mathbf{B}^{B}\mathbf{Y}^{BA}, \quad \text{and} \quad \mathbf{F}^{BB} = \mathbf{B}^{B}\mathbf{Y}^{BB}.$$

	Intermediate use			Final use			Inventories	Total
	Α	В	RoW	A	В	RoW		
Industries, A	\mathbf{Z}^{AA}	\mathbf{Z}^{AB}	\mathbf{Z}^{AR}	\mathbf{F}^{AA}	\mathbf{F}^{AB}	\mathbf{F}^{AR}	$\tilde{\mathbf{s}}^{A}$	\mathbf{x}^{A}
Industries, B	\mathbf{Z}^{BA}	\mathbf{Z}^{BB}	\mathbf{Z}^{BR}	\mathbf{F}^{BA}	\mathbf{F}^{BB}	\mathbf{F}^{BR}	$\tilde{\mathbf{s}}^{B}$	\mathbf{x}^{B}
Industries, RoW	\mathbf{Z}^{RA}	\mathbf{Z}^{RB}	\mathbf{Z}^{RR}	\mathbf{F}^{RA}	\mathbf{F}^{RB}	\mathbf{F}^{RR}		\mathbf{x}^{R}
Value added	$(\mathbf{w}^A)'$	$(\mathbf{w}^B)'$	$(\mathbf{w}^R)'$					
Int. transp. marg.	$(\mathbf{tm}_{\mathbf{U}}^{A})'$	$(\mathbf{tm}_{\mathbf{U}}^{B})'$	0	$(\mathbf{tm}_{\mathbf{Y}}^{A})'$	$(\mathbf{tm}_{\mathbf{Y}}^{B})'$	0	0	tm
Output	$(\mathbf{x}^A)'$	$(\mathbf{x}^B)'$	$(\mathbf{x}^R)'$					

The vectors for exports to RoW and changes in inventories at the industry level are obtained as

$$\tilde{\mathbf{e}}^{AR} = \mathbf{B}^A \mathbf{e}^{AR}, \quad \tilde{\mathbf{e}}^{BR} = \mathbf{B}^B \mathbf{e}^{BR}, \quad \tilde{\mathbf{s}}^A = \mathbf{B}^A \mathbf{s}^A, \quad \text{and} \quad \tilde{\mathbf{s}}^B = \mathbf{B}^B \mathbf{s}^B.$$

For the construction of WIOTs from world SUTs, one step was necessary that is not required when national IOTs are constructed. In the use tables in the international SUTs, information about imports (by product) from RoW is contained in \mathbf{U}^{RA} and \mathbf{U}^{RB} . To transform these product-by-industry matrices into industry-by-industry matrices, a similar transformation as \mathbf{B}^A and \mathbf{B}^B was needed for RoW. It was estimated by constructing a condensed supply table for the 40 WIOD countries (i.e. $\mathbf{V}^R = \mathbf{V}^A + \mathbf{V}^B$ for the two-country example of Figures 4 and 5) and dividing its cells by aggregated product output (i.e. $(\tilde{\mathbf{q}}^R)' = \iota' \mathbf{V}^R)$, which yields $\mathbf{B}^R = \mathbf{V}^R (\hat{\mathbf{q}}^R)^{-1}$. This procedure should yield a sensible estimate of the true transformation matrix if the distribution of products over supplying industries in RoW is rather similar to that of the countries covered in the WIOD. The estimates in Figure 5 are given by

$$\mathbf{Z}^{RA} = \mathbf{B}^{R}\mathbf{U}^{RA}, \quad \mathbf{Z}^{RB} = \mathbf{B}^{R}\mathbf{U}^{RB}, \quad \mathbf{F}^{RA} = \mathbf{B}^{R}\mathbf{Y}^{RA}, \text{ and } \mathbf{F}^{RB} = \mathbf{B}^{R}\mathbf{Y}^{RB}$$

The column $\tilde{\mathbf{e}}^{AR}$ with exports of country *A* to RoW was split into exports for intermediate use (\mathbf{Z}^{AR}) and for final use (\mathbf{F}^{AR}). For this, we used the average export shares to these use categories by developed economies to developing economies included in the WIOD. The developing economies considered are Brazil, Russia, India, China, Indonesia, and Mexico (BRICIM). For example, the average export shares of Austria's agricultural sector to the BRICIM across intermediate use industries and final use categories was used to split up the Austrian agricultural exports to RoW. This was done cell-wise (if the exporting country was a BRICIM country, it was excluded from the BRICIM average for these particular cells). The same approach resulted in the matrices \mathbf{Z}^{BR} and \mathbf{F}^{BR} in Figure 5.

Finally, the domestic deliveries in RoW (i.e. \mathbf{Z}^{RR} and \mathbf{F}^{RR}) have been estimated, based on data collected from the UN National Accounts. First, value-added data by economic activity and final demand category were summed for all countries not included in the WIOD to arrive at estimates of GDP by broad sectors and final demand categories for RoW. Gross output levels were obtained by applying the industry-specific average ratios of gross output to value added for developing economies in the WIOD (BRICIM). To split the broad manufacturing sector in the UN National Accounts into the considerably more disaggregated manufacturing industries in the WIOD, the average shares by the industry from the UNIDO industrial statistics for all countries not included in the WIOD for the period from 1995 to 2009 were used. Next, initial estimates of the domestic intermediate use block and the domestic final demand block have been obtained as weighted averages shares from the BRICIM countries. Jiang et al. (2012) arrived at the conclusion that such an 'averaging coefficients' procedure performs relatively well if unknown tables can be estimated by using data from other countries or regions for the same time period. Finally, to arrive at a balanced table, for which the column sums equal the row sums for RoW, we applied the RAS algorithm. To this end, we only RASsed the domestic block of RoW in the WIOTs, which implies that exports from RoW remained unchanged. The initial values fed into the algorithm were the input coefficients from the BRICIM countries, whereas the row and column totals were given by the externally provided data based on the UN National Accounts and UNIDO industrial statistics as described.

The WIOD database also contains WIOTs of the pyp. These tables have been obtained by applying row-wise deflation using industry output deflators, as published in the National Accounts. It should be stressed that the cells in the value-added row of the WIOTs in pyp have been obtained by double deflation, as residuals (the values of which ensure that the tables in pyp are balanced). NSIs derive value added in pyp (and therefore GDP in pyp) using much more detailed price information. Hence, the value-added figures in the WIOTs in pyp are not necessarily equal to those published by NSIs.²⁰ For value-added figures in constant prices as published by the NSIs, one is referred to the value-added volume indices in WIOD's socio-economic accounts. Dietzenbacher and Hoen (1998) proposed a method in which such externally constructed data in pyp can be used to arrive at an IOT in pyp, but this method is not suitable to address the problem at hand. It supposes that data on gross output and final demand in pyp are available at the level of industries. Such data are not available from National Accounts or any other source.

6. THE CONTENTS OF THE WIOD

The WIOD covers 27 EU countries and 13 other major countries in the world and provides annual data for the period from 1995 to 2009. The countries are listed in Table 2. They are selected on the basis of the quality and the public availability of their data in combination with their economic importance (together they account for approximately 85% of world GDP). All SUTs are of the format 35 industries by 59 products and all IOTs are 35 industries by 35 industries. The product and industry classifications are given in Tables A1 and A2, respectively, in a separate supplementary document that is accessible through the website of this journal. All data are downloadable at http://www.wiod.org/database/index.htm. The database includes the following information.

 $^{^{20}}$ For the RoW, this approach could not be used because industry output deflators are not systematically available for major countries that belong to RoW. Therefore, all cells in the RoW blocks have been deflated by the US GDP deflator. This approximation would be close to what would have been obtained with proper deflators if the exchange rates between the RoW currencies and the US dollar would be completely flexible, and the – in theoretical work popular – law of one price would hold. Both conditions are clearly not met in reality, but it is the best that can be attained, given data availability. Updates of the WIOD might contain WIOTs in pyp in which published value-added figures are combined with published industry gross output deflators, and where a biproportional method is used to yield balanced tables. More analysis is needed, however, to see whether such an approach yields empirically sensible results.

Euro-zone		Non-Euro EU	NAFTA	China	East Asia	BRIIAT
Austria Belgium Cyprus Estonia Finland France Germany Greece Ireland	Italy Luxembourg Malta Netherlands Portugal Slovakia Slovenia Spain	Bulgaria Czech Rep. Denmark Hungary Latvia Lithuania Poland Romania Sweden UK	Canada Mexico USA	China	Japan Korea Taiwan	Brazil Russia India Indonesia Australia Turkey

TABLE 2. WIOD countries and the regional aggregation adopted for the interregional WIOTS.

World tables

- International SUTs in current prices and pyp, with use split into domestically produced products and imported products by country of origin.
- WIOTs in current prices and pyp.
- Interregional IOTs for six regions (see Table A1 for the regional aggregations).

National tables

- National SUTs in current prices and pyp
- National IOTs in current prices

Socio-economic accounts (for details, see Erumban et al., 2012a)

- Industry output and value added, both in current and constant prices (indexed with 1995 as 100).
- Capital stock and investments.
- Wages and employment (in hours) by skill type, distinguishing low-, medium- and high-skilled workers.

Environmental accounts (for details, see Genty et al., 2012)

- Gross energy use by sector (i.e. industries and households) and energy commodity.
- Emission relevant energy use by sector and energy commodity.
- CO₂ emissions modelled by sector and energy commodity.
- Emissions to air by sector and pollutant.
- Land use, materials use and water use by type and sector.

7. CONCLUDING REMARKS

This paper describes the construction of the core of the WIOD: a time series of industryby-industry IOTs for 40 countries plus the RoW for the period 1995–2009. Starting from a set of underlying principles, among which reconciliation with time series from National

E. DIETZENBACHER et al.

Accounts and the use of official and public data are key), the construction procedure has been explained on a step-by-step basis. Some weaknesses of the approach have been touched upon, but there are more detailed issues that require closer scrutiny in follow-up research. An example of such an issue is international tourism. NAS appeared to record this part of international trade in very different ways. Given time limitations, the decision was made not to pursue this further, and to tackle problems that are relevant for almost all countries (such as differences in price concepts and treatment of re-exports and imports for processing activities). Nevertheless, data improvements for countries that depend relatively heavily on tourism seem possible, and more examples of opportunities for further work could be given.

In the longer run, continuing developments in the international statistical arena might lead to fundamental improvements regarding the data on which WIOD-like WIOTs could be constructed. First, the country coverage could most likely be improved after the Asian Development Bank launched an effort to assist NSIs of export-oriented countries such as Malaysia, the Philippines, Thailand and Vietnam in starting to construct SUTs. Second, the recent OECD–WTO 'Made in the World' Initiative has already led to plans regarding the systematic construction of data material that would explicitly distinguish between production structures of the parts of industries that export and of those parts that only deliver to domestic purchasers. Pioneering work by Chen et al. (2008) for China has shown that the implications of such a distinction for 'trade in value added' figures are substantial (see also Koopman et al., 2010, Johnson and Noguera, 2012).

A final long-term opportunity to improve on the current state of affairs is related to trade in services. The increasing importance of trade in services has led the United Nations Statistics Division to introduce a separate section on services trade in its trade statistics commission. Hopefully, this will lead to a much better coverage of bilateral services trade in a product classification, together with a distinction into end-use categories.

In the last section, we listed the contents of the satellite accounts that accompany the IOTs. These satellite accounts contain industry-level indicators regarding the inputs of production factors (such as capital and various types of labour), energy use and indicators of pollution due to industrial activity (such as emissions of greenhouse gases). These data enabled us already to come up with answers to the first question posed in the introduction. Between 1995 and 2007, the number of Chinese jobs that depended on consumption and investment demand in foreign countries increased from about 110 million to about 180 million (Los et al., 2012). Of these 180 million workers, actually more than 60% were not working in the manufacturing industries although Chinese exports are mainly manufacturing goods. This shows the importance of the indirect effects to which input-output analysis pays much attention. In other work, we found that fragmentation has led to a new division of tasks between advanced and emerging economies, with the former increasingly specializing in capital and high-skill intensive production activities (Timmer et al., 2012). We hope and think that combining data from these satellite accounts and the WIOTs in the WIOD will lead to many more studies that empirically examine the socio-economic and environmental consequences of globalization.

Acknowledgements

The WIOD project was funded by the European Commission, Research Directorate General as part of the seventh Framework Programme, Theme 8: Socio-Economic Sciences and

Humanities. Grant Agreement no: 225 281. Parts of this paper have been used also in Dietzenbacher et al. (forthcoming). We benefited greatly from discussions on this topic with many colleagues over the past years, which are too numerous to thank individually. Also thanks to Arnold Tukker and the three anonymous referees for providing useful comments and suggestions on this paper.

References

Andrew, R. and G.P. Peters (2013) A Multi-Region Input-Output Table Based on the Global Trade Analysis Project Database (GTAP-MRIO). *Economic Systems Research*, 25, 99–121.

- Baldwin, R. (2006) Globalisation: The Great Unbundling(s). Helsinki, Report for the Economic Council of Finland, available at: http://graduateinstitute.ch/webdav/site/ctei/shared/CTEI/Baldwin/Publications/Chapters/ Globalization/Baldwin_06-09-20.pdf
- Chen, X., L. Cheng, K.C. Fung and L. Lau (2008) The Estimation of Domestic Value-Added and Employment Induced by Exports: An Application to Chinese Exports to the United States. In: Y.W. Cheung and K.Y. Wong (eds.) China and Asia: Economic and Financial Interactions. Routledge, Abingdon, UK, Routledge, 64–82.
- Costinot, A., J. Vogel and S. Wang (forthcoming) An Elementary Theory of Global Supply Chains. Review of Economic Studies. doi:10.1093/restud/rds023
- Dietzenbacher, E. and A.R. Hoen (1998) Deflation of Input-Output Tables from the User's Point of View: A Heuristic Approach. *Review of Income and Wealth*, 44, 111–122.
- Dietzenbacher, E., B. Los and M.P. Timmer (forthcoming) The World Input-Output Tables in the WIOD Database. In: J. Murray and M. Lenzen (eds.) *The Sustainability Practitioner's Guide to Multi-Regional Input-Output Analysis*. Champaign, IL, Common Ground Publishing.
- Erumban, A., R. Gouma, G. de Vries, K. de Vries and M.P. Timmer (2012a) WIOD Socio-Economic Accounts (SEA): Sources and Methods, available at: http://www.wiod.org/publications/source_docs/SEA_Sources.pdf
- Erumban, A., R. Gouma, G. de Vries, K. de Vries and M.P. Timmer (2012b) Sources for National Supply and Use Table Input Files, available at: http://www.wiod.org/publications/source_docs/SUT_Input_Sources.pdf
- Eurostat. (2008) Eurostat Manual of Supply, Use and Input-Output Tables. Luxembourg, Eurostat.
- Foster, N., J. Francois, O. Pindyuk, J. Poschl and R. Stehrer (2012) WIOD International Trade in Goods and Services Data. WIOD Deliverable D2.3.
- Francois, J. and B. Hoekman (2010) Services Trade and Policy. Journal of Economic Literature, 48, 642–692.

Genty, A., I. Arto and F. Neuwahl (2012) Final Database of Environmental Satellite Accounts: Technical Report on their Compilation, available at: http://www.wiod.org/publications/source_docs/Environmental_Sources.pdf Grossman, G. and E. Rossi-Hansberg (2008) Trading Tasks: A Simple Theory of Offshoring. American Economic

- *Review*, 98, 1978–1997. Guo, D., C. Webb and N. Yamano (2009) Towards Harmonised Bilateral Trade Data for Inter-country Input-Output
- Guo, D., C. webb and N. ramano (2009) Towards Harmonised Bilateral Trade Data for Inter-country input-Output Analyses: Statistical Issues. Paris, OECD STI Working Papers 2009/4, OECD, Directorate for Science, Technology and Industry.
- ISWGNA (1994) System of National Accounts 1993. Brussels/Luxembourg, New York, Paris, Washington: D.C. Commission of the European Communities-Eurostat, International Monetary Fund, OECD, United Nations.
- ISWGNA (2009) System of National Accounts 2008. Brussels/Luxembourg, New York, Paris, Washington: D.C. Commission of the European Communities-Eurostat, International Monetary Fund, OECD, United Nations.
- Jackson, R.W. and A.T. Murray (2004) Alternative Input-Output Matrix Updating Formulas. Economic Systems Research, 16, 135–148.
- Jiang, X., E. Dietzenbacher and B. Los (2012) Improved Estimation of Regional Input-Output Tables Using Cross-Regional Methods. *Regional Studies*, 46, 621–637.
- Johnson, R.C. and G. Noguera (2012) Accounting for Intermediates: Production Sharing and Trade in Value Added. Journal of International Economics, 86, 224–236.
- Koopman, R., W. Powers, Z. Wang and S.-J. Wei (2010) Give Credit where Credit is Due: Tracing Value Added in Global Production Chains. NBER Working Paper 16426, Cambridge, MA.
- Lahr, M.L. and L. de Mesnard (2004) Biproportional Techniques in Input-Output Analysis: Table Updating and Structural Analysis. *Economic Systems Research*, 16, 115–134.
- Lenzen, M. (2011) Aggregation vs. Disaggregation in Input-Output Analysis of the Environment. Economic Systems Research, 23, 73–89.
- Lenzen, M., B. Gallego and R. Wood (2007) Some Comments on the GRAS Method. Economic Systems Research, 19, 461–465.
- Lenzen, M., B. Gallego and R. Wood (2009) Matrix Balancing Under Conflicting Information. *Economic Systems Research*, 21, 23–44.
- Lenzen, M., D. Moran, K. Kanemoto and A. Geschke (2013) Building EORA: A Global Multi-Region Input-Output Database at High Country and Sector Resolution. *Economic Systems Research*, 25, 20–49.

- Los, B., M.P. Timmer and G.J. de Vries (2012) China and the World Economy: A Global Value Chain Perspective on Exports, Incomes and Jobs. GGDC Research Memorandum 128, University of Groningen.
- Meng, B., Y. Zhang and S. Inomata (2013). Compilation and Applications of IDE-JETRO's International Input-Output Tables. *Economic Systems Research*, 25, 122–142.
- Miller, R.E. and P.D. Blair (2009) Input-Output Analysis: Foundations and Extensions. 2nd ed. Cambridge, Cambridge University Press.
- Minguez, R., J. Oosterhaven and F. Escobedo (2009) Cell-Corrected RAS Method (CRAS) for Updating or Regionalizing an Input-Output Matrix. *Journal of Regional Science*, 49, 329–348.
- OECD–WTO. (2012) Measuring Trade in Value-Added: An OECD-WTO Joint Initiative, available at: http://www. oecd.org/sti/industryandglobalisation/measuringtradeinvalue-addedanoecd-wtojointinitiative.htm.
- O'Mahony, M. and M.P. Timmer (2009) Output, Input and Productivity Measures at the Industry Level: The EU KLEMS Database. *Economic Journal*, 119, F374–F403.
- Oosterhaven, J., D. Stelder and S. Inomata (2008) Estimating International Interindustry Linkages: Non-Survey Simulations of the Asian-Pacific Economy. *Economic Systems Research*, 20, 395–414.
- Peters, G.P., R. Andrew and J. Lennox (2011) Constructing an Environmentally Extended 1132 Multi-Regional Input-Output Table Using the GTAP Database. *Economic Systems 1133 Research*, 23, 131–152.
- Puzzello, L. (2012) A Proportionality Assumption and Measurement Biases in the Factor Content of Trade. Journal of International Economics, 87, 105–111.
- Rueda-Cantuche, J.M. and T. ten Raa (2009) The Choice of Model in the Construction of Industry Coefficients Matrices. *Economic Systems Research*, 21, 363–376.
- Streicher, G. (2012) Estimation of Transport Costs for the WIOD Project (Paper prepared for the final report of the WIOD project).
- Streicher, G. and R. Stehrer (2012) Wither Panama? Constructing a Consistent and Balanced World SUT System Including International Trade and Transport Margins. WIOD Working Paper Nr. 13, available at: www.wiod.org.
- Temurshoev, U. and M.P. Timmer (2011) Joint Estimation of Supply and use Tables. *Papers in Regional Science*, 90, 863–882.
- Temurshoev, U., C. Webb and N. Yamano (2011) Projection of Supply and Use Tables: Methods and their Empirical Assessment. *Economic Systems Research*, 23, 91–123.
- The Economist. (2012) The Retreat of the Monster Surplus, 26 May, available at: http://www.economist.com/node/ 21555767.
- Timmer, M.P. (ed., 2012) The World Input-Output Database (WIOD): Contents, Sources and Methods. WIOD working paper nr. 10, available at: http://www.wiod.org.
- Timmer, M.P., B. Los, R. Stehrer and G.J. de Vries. (2012) Fragmentation, Income and Jobs. An analysis of European competitiveness, GGDC Research memorandum Nr. 129, University of Groningen, The Netherlands.
- Tukker, A., A. de Koning, R. Wood, T. Hawkins, S. Lutter, J. Acosta, J.M. Rueda Cantuche, M. Bouwmeester, J. Oosterhaven, T. Drosdowski and J. Kuenen (2013) EXIOPOL – Development and Illustrative Analyses of a Detailed Global MR EE SUT/IOT. *Economic Systems Research*, 25, 50–70.
- Wang, Z., M. Tsigas, J. Mora, X. Li and D. Xu (2010) A Time Series Database for Global Trade, Production, and Consumption Linkages (Paper presented at the GTAP Conference on Global Economic Analysis, Penang, Malaysia, June 2010).
- Wiebe, K.S., M. Bruckner, S. Giljum and C. Lutz (2012) Calculating Energy-related CO₂ Emissions Embodied in International Trade Using a Global Input-Output Model. *Economic Systems Research*, 24, 113–139.
- Wiedmann, T., H.C. Wilting, M. Lenzen, S. Lutter and V. Palm (2011) Quo Vadis MRIO? Methodological, Data and Institutional Requirements for Multi-Region Input-Output Analysis. *Ecological Economics*, 70, 1937–1945.